Author's response to reviews

Title: Impact of case management by advanced practice nurses in primary care on unplanned hospital admissions: a controlled intervention study

Authors:

Dyfed Huws (dyfedhuws@yahoo.co.uk)
Deborah Cashmore (debbie.cashmore@swansealhb.wales.nhs.uk)
Robert Newcombe (newcombe@Cardiff.ac.uk)
Catherine Roberts (Catherine.Roberts@swansealhb.wales.nhs.uk)
Judith Vincent (judith.vincent@swansealhb.wales.nhs.uk)
Glyn Elwyn (elwyn@cardiff.ac.uk)

Version: 2 Date: 10 September 2007

Author's response to reviews:

Background and methods have been revised in abstract.